Author's response to reviews

Title: A New Human Heart Vessel Identification, Segmentation and 3D Reconstruction Mechanism

Authors:

Aqeel A. Al-Surmi (ageel_ye@yahoo.com)
Rahmita W. Rahmat (rahmita@upm.edu.my)
Ramlan B. Mahmod (ramlan@upm.edu.my)
Fatimah B. Khalid (fatimahk@upm.edu.my)
Zamrin Dimon (zamrin7680@salam.uitm.edu.my)

Version: 2
Date: 16 May 2014

Author's response to reviews: see over
Cover Letter

Dear Sir\Madam,

Regarding the paper titled:” A New Human Heart Vessel Identification, Segmentation and 3D Reconstruction Mechanism”. We would like to inform you that to the best of our knowledge, this work does not currently exist in print or otherwise and currently, the paper has not been submitted for publication anywhere, and it will not be submitted until a decision has been made.

Furthermore, the author would like to express their grateful to your estimated Journal to support and serve the knowledge’s.

Best regards,
The Authors of this paper are:
   1. Aqeel A. Al-Surmi a (Corresponding Author)
   2. Rahmita Wirza a
   3. Ramlan Mahmod a
   4. Fatimah Khalid a
   5. Dr. Mohd Zamrin Dimon b

a Department of Multimedia, Faculty of Computer Science and Information Technology, University Putra Malaysia, Malaysia,

b Cardiothoracic Unit, Surgical Cluster, Faculty of Medicine, Universiti Teknologi MARA, Malaysia.

Universiti Putra Malaysia
43400 Serdang, Selangor
Tel. : 603-89471704
Fax : 603-8946576
E-mail: aqeel_ye@yahoo.com